Author’s response to reviews

Title: Renal cell carcinoma with a tumor thrombus in the ureter; a case report.

Authors:

Osamu Fujita (urosamu@gmail.com)
Koichiro Wada (gmd17055@s.okadai.jp)
Tomoya Yamasaki (kingsusa0416@yahoo.co.jp)
Daisuke Manabe (dai-sato@ic4.so-net.ne.jp)
Katsuji Takeda (k-takeda@chp-kagawa.jp)
Satoko Nakamura (s-nakamura@chp-kagawa.jp)

Version: 2 Date: 4 May 2011

Author’s response to reviews:

Dear editors

I emphasized this theme of article, as tumor thrombus in "whole" the renal pelvis and ureter.
I omitted extra information of the clinical findings.
I shortened discussion and squeezed the point as rareness and difficulty of preoperative diagnosis.
I checked and corrected spelling of some words.